

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Vinh Xuan Bui; Chu, Thi Kim Huong

Article

Do foreign shareholders improve corporate earnings quality in emerging markets? Evidence from Vietnam

Cogent Economics & Finance

Provided in Cooperation with:

Taylor & Francis Group

Suggested Citation: Vinh Xuan Bui; Chu, Thi Kim Huong (2019) : Do foreign shareholders improve corporate earnings quality in emerging markets? Evidence from Vietnam, Cogent Economics & Finance, ISSN 2332-2039, Taylor & Francis, Abingdon, Vol. 7, Iss. 1, pp. 1-22, https://doi.org/10.1080/23322039.2019.1698940

This Version is available at: https://hdl.handle.net/10419/270706

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



https://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU





Cogent Economics & Finance

ISSN: (Print) 2332-2039 (Online) Journal homepage: https://www.tandfonline.com/loi/oaef20

Do foreign shareholders improve corporate earnings quality in emerging markets? Evidence from Vietnam

Xuan Vinh Vo & Thi Kim Huong Chu |

To cite this article: Xuan Vinh Vo & Thi Kim Huong Chu | (2019) Do foreign shareholders improve corporate earnings quality in emerging markets? Evidence from Vietnam, Cogent Economics & Finance, 7:1, 1698940, DOI: <u>10.1080/23322039.2019.1698940</u>

To link to this article: <u>https://doi.org/10.1080/23322039.2019.1698940</u>

© 2020 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license.



6

Published online: 08 Jan 2020.

|--|

Submit your article to this journal 🕝

Article views: 1625



View related articles 🗹

🕨 View Crossmark data 🗹



Citing articles: 10 View citing articles 🕝



Received: 23 September 2019 Accepted: 21 November 2019 First Published: 30 November 2019

*Corresponding author: Xuan Vinh Vo, Institute of Business Research and CFVG Ho Chi Minh City, University of Economics Ho Chi Minh City, 59C Nguyen Dinh Chieu Street, District 3, Ho Chi Minh City, Vietnam E-mail: vxvinh@ydhoo.com

Reviewing editor: David McMillan, University of Stirling, Stirling, UK

Additional information is available at the end of the article

FINANCIAL ECONOMICS | RESEARCH ARTICLE Do foreign shareholders improve corporate earnings quality in emerging markets? Evidence from Vietnam

🔆 cogent

economics & finance

Xuan Vinh Vo^{1*} and Thi Kim Huong Chu²

Abstract: Employing a panel dataset of Vietnamese non-financial listed firms, we find that firms with greater foreign shareholdings are aligned with higher quality of financial disclosure. More specially, we find that greater foreign shareholdings are associated with (i) lower earnings management; (ii) more persistent earnings; and (iii) higher informative earnings. On the ground that foreign investors in Vietnam equity market are dominated by institutional investors, the finding from this study supports the spillover hypothesis. This suggests that foreign institutional shareholders with extensive management skills might have ability to assist their invested firms in improving quality of reported earnings.

Subjects: Quantitative Finance; Corporate Finance; Business, Management and Accounting

Keywords: earnings quality; foreign ownership; spillover hypothesis; Vietnam stock market JEL classifications: G23; G32; M41; M43; M45

1. Introduction

The association between ownership structure and corporate governance has received great attention for years by financial market participants, researchers, and policy makers. A considerable amount of papers in finance literature focuses on analyzing this relation from different perspectives, including agency cost, fraud, and audit quality (Chen, Firth, Gao, & Rui, 2006; McKnight & Weir, 2009; Singh & Davidson III, 2003; Xu & Wang, 1999). In the last two decades, there is an increasing of corporate governance studies which recognize ownership structure as a mechanism for improving the quality of corporate earnings (Aydin, Sayim, & Yalama, 2007; Ben-Nasr, Boubakri, & Cosset, 2009; de Sousa & Galdi, 2016; Jung & Kwon, 2002; Katz, 2009; Klai & Omri, 2011; Mehrani, Moradi, & Eskandar, 2017; Sánchez-Ballesta & García-Meca, 2007).

ABOUT THE AUTHORS

Xuan Vinh Vo is from the Institute of Business Research, University of Economics Ho Chi Minh City, Vietnam and CFVG Ho Chi Minh City, University of Economics Ho Chi Minh City, Vietnam. His expertise is in the areas of banking and finance.

Thi Kim Huong Chu is a research fellow from the Institute of Business Research, University of Economics Ho Chi Minh City. Her research areas are accounting and finance.

PUBLIC INTEREST STATEMENT

In this paper, we employ a panel dataset of Vietnamese non-financial listed firms to explore the link between foreign shareholdings and earnings quality. We report that firms with greater foreign shareholdings are aligned with higher quality of financial disclosure. More specifically, we find that greater foreign shareholdings are associated with (i) lower earnings management; (ii) more persistent earnings; and (iii) higher informative earnings. Our finding supports the spillover hypothesis. This suggests that foreign shareholders with extensive management skills might have ability to assist their invested firms in improving quality of reported earnings.





 ${}^{\odot}$ 2020 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license.

Recent financial globalization trend opens more space for international capital flow in emerging markets. In response to this trend, the presence of foreign investors is increased in developing countries (Huang & Shiu, 2009). Many authors concur that international investors face several challenges, such as asymmetry information, difficulty in communication, and difference in business culture when conducting investment in newly established markets (Cao, Du, & Hansen, 2017; Karolyi & Stulz, 2003). In addition, emerging markets are featured by weak protection for minority shareholder (La Porta, Lopez-de-Silanes, Shleifer, & Vishny, 2000). These difficulties might equip foreign investors with stronger incentives to protect their investments. As a result, it is important to address a natural question whether foreign shareholders influence the corporate governance of domestic companies.

While there are a few studies investigating the influence of foreign ownership in emerging markets (Baba, 2009; Cao et al., 2017; Desender, Aguilera, Lópezpuertas-Lamy, & Crespi, 2014; Kang, Sul, & Kim, 2010), the role of foreign shareholders in improving earnings quality in Vietnam stock market remains unexplored in the current literature. To fill this gap, this paper examines the relationship between foreign ownership and firm's earnings quality in the context of Vietnam, an important emerging market.

A number of published studies provide evidence that foreign investors have a strong influence on stock market activity in Vietnam stock market. For example, foreign investors play an important role in stabilizing the stock price volatility (Vo, 2015) and containing the risk-taking activities in domestic companies (Vo, 2016). In line with previous studies, we further contribute to the literature on the impact of foreign shareholders on earnings quality. We aim to provide an extensive analysis concerning the role of foreign investors in Vietnam equity market. In particular, we address the question whether foreign shareholders influence the quality of earnings using a sample of non-financial companies listed on the Ho Chi Minh City Stock Exchange over an extended period.

Earnings quality has been an important focus in several studies (Dechow & Dichev, 2002; Dechow, Ge, & Schrand, 2010; Fan & Wong, 2002; He, Ng, Zaiats, & Zhang, 2017; Hribar & Craig Nichols, 2007; Schipper & Vincent, 2003). However, there is no universal and single definition to measure for earnings quality. These previous studies highlight the importance of appropriate proxies for earnings quality. Consequently, previous studies use a wide range of earnings quality concepts and measures, including both accounting measures (earnings persistence, predictability timeliness, earnings management, and earnings smoothness) and market measures (earnings quality is a multidimensional concept and researchers typically use different proxies of earnings quality in their studies (Fan & Wong, 2002; Francis, LaFond, Olsson, & Schipper, 2004; Myers, Myers, & Omer, 2003; Richardson, Sloan, Soliman, & Tuna, 2001; Sloan, 1996; Wild, 1996).

Investigating earnings quality by using proxies containing one-dimensional angle may induce inadequate conclusions. To address this shortcoming, this paper analyzes earnings quality from two perspectives, including both accounting-based earnings quality and market-based earnings quality. Moreover, as argued by Dechow et al. (2010), the common measures of accounting-based earnings quality are earnings management and earnings persistence, whereas earnings informativeness is the most powerful proxy for market-based earnings quality. In light of above arguments, this research will employ two measurements of earnings quality, namely accounting measure (earnings management and earnings persistence) and market measure (earnings informativeness).

The theoretical literature offers two competing theories to explain the impact of foreign stakeholders on the quality of corporate accounting earnings. The former refers to the knowledge spillover hypothesis. Studies supporting this hypothesis argue that when competent international investors make investment in domestic companies, they have incentives to demand better accounting standards, do greater monitoring to limit the earnings management of firms and generate the high quality of reported earnings (Abor & Biekpe, 2007; Aydin et al., 2007; Ben-Nasr, Boubakri, & Cosset, 2015; Choi, Lam, Sami, & Zhou, 2013; Firth, Fung, & Rui, 2007; Guo, Huang, Zhang, & Zhou, 2015). The latter relies on the information asymmetry hypothesis (Dvořák, 2005; Klai & Omri, 2011; Xiao, Yang, & Chow, 2004). This hypothesis argues that the constraint of geographic distance and barrier of language limits foreign investors to manage the firm's accounting operation. Consequently, they have less power to oversee the local firms' governance.

Drawing upon two strands of literature on the relationship between ownership of foreign investors and earnings quality, this study attempts to explore the nature of foreign stakeholder's role in enhancing the quality of financial reported earnings using a sample of Vietnamese listed firms. More specially, we analyze the association between foreign ownership and earnings quality in Vietnam stock market. We employ three main attributes of earnings quality, namely earnings persistence, earnings management, and earnings informativeness.

Our investigation of the association between foreign shareholdings and quality of reported earnings is important in the context of Vietnam because of various angles. Firstly, in the last few years, with continuous efforts to liberalize and internationalize domestic capital markets, Vietnam promotes economic integration and embarks on several economic policy reforms. Generally, many of these reforms involve elimination of investment barriers on foreign investors. Easing restrictions on foreign ownership in domestic companies is the most important subset for economic reforms. Particularly, a significant reform is that Vietnamese government relaxed the foreign ownership caps in non-banking companies in 2015. This law allows foreigners to own up to 100% of the local firm, which is a significant increase from the previous cap of 49%. This reform naturally provides us a unique opportunity to initiate a comprehensive research of these issues within an emerging market setting.

Secondly, our study is important because the inflow of foreign portfolio investment in Vietnam increases significantly in the last decades. Particularly, data from the World Bank show that foreign portfolio investment inflows into Vietnam significantly increase during the period from 2007 to 2015.¹ This outcome is an important achievement resulting from the political stability and high economic growth in Vietnam over the past decades.

Thirdly, the current study is important in the process of privatization in emerging economies. The shift in government policies which encourage the transformation of corporate ownership structure from state-owned companies into privatized firms brings foreign investors more pace to raise their stake. This is evidenced by the numbers of privatized companies are 289 in 2015 from 95 firms in 2012.²

The analysis for the case of a specific country enables us to adopt a more direct test of the impact of foreign investors on financial reporting quality. The difference in institutional setting and legal framework between well-established markets and newly established markets is documented in many previous studies (Fan & Wong, 2002; Khalil & Ozkan, 2016; La Porta & Lopez-de-Silanes, 1998). In contrary to developed countries, developing countries are characterized by substantial ownership concentration and insufficient disciplines of corporate governance (World Bank, 2013). Moreover, in a unique institutional setting, the shareholders and managers might justify their behavior and discretion according to the nature of this particular economy (Cuomo, Mallin, & Zattoni, 2016; Haxhi & Aguilera, 2017).

This research uses a dataset covering 245 non-financial firms listing on the Ho Chi Minh City Stock Exchange over the period from 2007 to 2015. The current study demonstrates significant evidence that foreign shareholders serve as an active monitor in enhancing earnings quality of domestic companies. In other words, this study offers evidence of a positive correlation between foreign stakeholders and quality of reported earnings. More specifically, our results show that higher foreign shareholdings are associated with (i) lower earnings management, (ii) more earnings persistence, and (iii) higher informative earnings.

Our analysis offers several implications for investment practice and policy development. Primarily, the paper is important in the sense that emerging markets become more important for foreign investors and policy makers (Cao et al., 2017; Li & Giles, 2015). The finding of a positive effect of foreign shareholdings on local firm's corporate governance might provide valuable information to investors in formulating their investment strategy. Furthermore, although the legal act easing the limit on foreign ownership is promulgated, it does not provide detailed instructions for firms. Clearly, further guidelines are required because both listed companies and investors are still unclear about which specific industries and firms will be fully opened for foreign investors. Hence, the empirical findings in this paper are relevant for policy makers in improving transparency and developing policies. More importantly, removing restrictions on foreign portfolio investment is a priority subset in Vietnam's economic reforms.

The current research contributes to the finance literature in different dimensions. Clearly, the current study uses a dataset and the context of an important emerging market which has dissimilar features in comparison with the developed markets. This allows us to proclaim the critical role of the institutional setting in analyzing the relationship between foreign ownership and earnings quality. Additionally, the paper offers further insights to the current literature by providing evidence of a positive linkage between foreign holdings and corporate quality of earnings.

The structure of the paper is designed as follows. Section 2 presents salient features of foreign ownership in Vietnam stock market. Section 3 reviews related literature and finds there are divergent predictions on the relationship between foreign ownership and earnings quality. Section 4 introduces the sampling procedure and research methodology. Section 5 presents the empirical results and discussion of results. Finally, Section 7 offers some conclusions, insight implications, and suggestions for future research.

2. Stylized facts about foreign ownership in Vietnam stock market

The increased interest from foreign investors is in line with the equity market development in Vietnam. Figure 1 illustrates the foreign securities trading accounts in relative with the market capitalization over the period of 2010-2016. Apparently, the enlarged trading accounts of foreign investors are aligned with the improvement of market capitalization.

Figure 2 reports the proportion of foreign individuals and foreign institutions in total foreign trading value. Apparently, the foreign institutional investors account for over 90% in total value of foreign trading. This suggests that in Vietnam stock market, international institutions play a dominant role in foreign trading activities.

Table 1 shows the descriptive summary of foreign ownership in different industries following the Global Industry Classification Standard (GICS) in Vietnam stock market as of 31 December 2016. Total market capitalization of firms in the sample is presented in the second column of this table. We observe that Consumer Staples is the largest industry which accounts for 37.3% of the total market capitalization while Energy is the smallest industry with only 0.9% of the total market capitalization.

The fourth column presents market capitalization of firms with foreign ownership by industry. Although Consumer Staples is the largest industry in Vietnam stock market, foreign investors appear to prefer financial industry where foreign investors invest a large proportion of their holdings in financial stocks (29.3%). Consumer Staples and Real Estate also attract foreign attention with 27.7% and 12%, respectively. Notably, despite the actual proportion of foreign holdings in pharmaceutical firms is quite high (29.82%) in comparison with the total firms' average value

Figure 1. Foreign securities trading accounts on Vietnamese stock market.

Notes: This figure presents the number of foreign trading accounts over the period of 2010–2016. The bars represent the market capitalization of the Ho Chi Minh City Stock Exchange, expressed in million Vietnam Dong. This measure is reflected on the left scale. The line shows the number of international account holders, reflected on the right scale. Data are collected from Bloomberg and Vietnam Securities Depository.

Figure 2. Foreign trading on Vietnam equity market.

Notes: The stacked bar chart shows the proportion of trading value by foreign individuals and foreign institutions (in percentage) in 2016. The portion of foreign trading is reflected on the left scale. The gray bars represent the portion of foreign institutional trading value, the black bars illustrate the percentage of foreign individual trading value on the Ho Chi Minh City Stock Exchange (expressed in percentage). Data are provided by Fiinpro, a commercial data supplier in Vietnam.



Foreign trading



(14.06%), foreigners invest only 1.1% of market capitalization in Health Care industry. The possible explanation for this phenomenon is because there is a small number of Vietnamese pharmaceutical firms listing their shares on the exchange so that the weight of medicines companies in the foreign portfolio is relatively low.

Regarding the legal framework, the first legal provision controlling foreign ownership restriction in listed companies is introduced in 2009. This early law holds significant constraints for foreign market participants. Especially, this law stipulates that foreign investors are not permitted to own more than 49% of total outstanding shares of a public company. This legal framework, with the purpose of protecting local investors, however, becomes an obstacle for the development of domestic listed companies. This is also contrary to the trend of increased international economic integration. Some listed companies attempt to avoid this restriction by registering to delist in order to raise more foreign capital. Furthermore, restrictions on foreign ownership in public enterprises also create difficulties for foreign fund managers in rebalancing their portfolio. This restriction also prevents foreign investors from excessive trading which in turn reduces stock market liquidity.

Table 1. Foreig					
Industry	Listed firm	s in industry	Listed firms shareł	Average foreign	
	Number of firms	Mkt. Cap. (%)	Number of firms (%)	Mkt. Cap. (%)	ownership (%)
Consumer Discretionary	36	2,115 (2.9)	34 (94.4)	787 (4.7)	12.79
Consumer Staples	36	20,027 (37.3)	35 (97.2)	6,269 (27.7)	19.85
Energy	9	539 (0.9)	9 (100)	146 (0.7)	20.09
Financials	21	21,175 (27.7)	20 (95.2)	4,659 (29.3)	18.23
Health Care	9	813 (2.1)	9 (100)	354 (1.1)	29.82
Industrials	91	7,312 (9.3)	89 (97.8)	1,567 (10.1)	14.12
Information Technology	7	1,167 (3.2)	7 (100)	534 (1.6)	17.78
Materials	52	3,805 (6)	52 (100)	1,002 (5.3)	9.02
Real Estate	37	8,703 (6.8)	36 (97.2)	1,139 (12)	13.86
Utilities	19	6,549 (2)	19 (100)	332 (9.1)	10.37
Not Classified	4	44 (0.06)	4 (100)	0.58 (0.003)	0.85
All	321	72,250	314	16,790	14.06

Notes: This table presents the industry classification and foreign ownership by industry on the Ho Chi Minh City Stock Exchange as of 30 December 2016. Data are collected from Bloomberg. The total number of firms by industry is appeared in the first column and the number of firms with foreign presence is reported in the third column. The second and fourth column illustrate market capitalization (Mkt. Cap) of firms in total sample and firms with foreign ownership, respectively. The last column contains data of average foreign ownership in each industry. The parentheses consist of corresponding proportion of individual industry's value to the whole sample's value (in percentage).

The new regulatory framework relaxing foreign ownership caps is introduced in 2015 to attract more foreign capital inflows but still delay in terms of application. This provision states that if annual general shareholders meeting approves on foreign limit relaxation, foreign investors can increase their holdings to one hundred percent (100%) of a firm's outstanding shares. The new legal framework encourages public firms to enhance their business performance by approaching to foreign investment capital. Nonetheless, the law is note clear on which particular sectors or industries which are qualified to remove the limit on holdings of foreign investors.

With respect to corporate governance, Vietnam Corporate Governance Code is promulgated in 2007, marking a positive step in enhancing domestic firm's governance. This Code regulates the internal governance structures of a listed company, investor protection, and information transparency. Since the compulsory corporate governance regulations are applied, the privatization progress of state-ownership enterprises is strongly promoted (World Bank, 2013). In addition, this framework is given certain rights to shareholders, for example, approval of dividends, voting rights to elect the board of directors. Moreover, this legal work enables firm's shareholders to access both financial reports and non-financial information.

3. Related studies on the association between foreign ownership and earnings quality

In the existing earnings quality literature, investigation of the relation between foreign shareholdings and quality of accounting earnings generally leads to divergent and inconclusive results. More specially, some researchers report a positive relationship while others document a negative relationship between these two important variables. The former is commonly explained by the spillover hypothesis and the latter is supported by the information asymmetry hypothesis.

The spillover hypothesis predicts that foreign investors with strong management skills can enhance the quality of reporting earnings. For example, Firth et al. (2007) employ two proxies for earnings

quality, including the earnings response coefficients and discretionary accruals to report that the presence of foreign shareholders positively impacts the earnings quality. A possibility is that the foreign owners force local firm's managers to enhance the quality of financial reporting. Similarly, Abor and Biekpe (2007) find that foreign investors have a significantly positive influence on the quality of earnings in Ghanaian small and medium enterprises. In the same vein, utilizing dataset of firms listed on the Istanbul Stock Exchange, Aydin et al. (2007) conclude that foreign investors play important role in improving the financial performance of local firms. This study also provides evidence that foreign stakeholders improve company's performance better than domestic owners do.

More recently, Ben-Nasr et al. (2015) comprise the sample of firms from 45 countries to examine the influence of foreign shareholders on the quality of earnings. The authors conclude that greater level of international shareholdings is correlated with better earnings disclosure. Likewise, Guo et al. (2015) use a unique set of foreign ownership dataset of listed Japanese firms to explore whether foreign investors influence real earnings management. This study offers evidence that foreign ownership is negatively related to earnings management. In other words, foreign investors play the monitoring role in alleviating the earnings management activities of local firms.

On the contrary, a large volume of papers find a negative association between the level of foreign ownership and the degree of earnings quality. This finding could be explained by the informational asymmetry hypothesis. For example, Xiao et al. (2004) report that larger foreign holdings ratio is aligned with lower quality of reported earnings. This is because foreign investors have the constraint of higher information asymmetry in comparison with local owners. Equally, Klai and Omri (2011) utilize the data set of listed companies on the Tunis Stock Exchange throughout the period from 1997 to 2007 and document that the increased power of foreigners negatively affects financial reporting quality. Those authors suggest the possible explanation for the negative relationship between these two variables relying on the information asymmetry hypothesis. Especially, they state that foreign investors hardly control a firm's accounting operation because they face information disadvantages due to geographic distance, language barriers, and cultural differences.

4. Data and research methodology

4.1. Data

There are currently two main stock exchanges in Vietnam which are the Ho Chi Minh City Stock Exchange (HOSE) and the Hanoi Stock Exchange (HSX). However, the listing requirements and standards are different. The former is the largest bourse in Vietnam which is established in 2000 with 322 listed firms as of 31 December 2015. This bourse is also an attractive destination for foreign investors who invest in Vietnam. The latter is a smaller one which is established in 2005. We use the data for the largest exchange.

Our study utilizes the unbalanced sample of firms listed on the Ho Chi Minh City Stock Exchange. We include firms that have available data for the period from 2007 to 2015. At the end of 2015, there are 322 listed companies, including 57 financial institutions and 265 non-financial firms. The sample excludes 57 financial institutions because of their difference in financial statement presentation and business nature. Furthermore, the research eliminates 20 non-financial companies which we do not have data available in three years. Our final sample include 245 listed firms.

Both accounting and stock market trading data are obtained from Bloomberg database. Distribution of sample observation is presented in Table 2. Panel A shows the distribution of observation by year distribution of observation by industry is classified by GICS sectors.

Panel A: Distribution	on of observations by year
Year	No. of observation
2007	1,144
2008	1,469
2009	1,929
2010	2,388
2011	2,543
2012	2,572
2013	2,584
2014	2,569
2015	2,520
Total firm-year observations	19,718
Panel B: Distributio	n of observation by industry
GICs classification	No. of observation
Consumer Discretionary	2,558
Consumer Staples	2,568
Energy	713
Health Care	718
Industrials	5,708
Information Technology	478
Materials	3,396
Real Estate	491
Utilities	1,281
Not classified	1,807
Total firm-year observations	19,718

4.2. Methodology

Our proxies of earnings quality consist of accounting-based (earnings management, earnings persistence) and market-based (earnings informativeness). The benefits of approaching both perspectives are enhancing the robustness of the test and providing comprehensive results.

4.2.1. Earnings management

Previous studies suggest that the presence of earnings management results in poor quality of earnings (Deng, Li, & Liao, 2017; Hribar & Craig Nichols, 2007; Lo, 2008; Schipper & Vincent, 2003). Discretionary accruals are widely used as the common measure of earnings management (Ben-Nasr et al., 2009, 2015; Degeorge, Ding, Jeanjean, & Stolowy, 2013; Firth et al., 2007; Francis & Wang, 2008; Park & Park, 2004; Wang, 2006). Higher (smaller) discretionary accruals translate into greater (less) earnings management.

We perform two-step procedure to estimate the influence of foreign shareholdings on earnings management. We first conduct the total accruals regression to estimate discretionary accruals. We then estimate a regression between foreign ownership and discretionary accruals.

Specially, in the first step, we employ the measurement of discretionary accruals using the approach of Dechow and Dichev (2002) and Degeorge et al. (2013):

$$\frac{CAC_{it}}{TA_{it}} = \theta_0 + \theta_1 \frac{CFO_{it-1}}{TA_{it}} + \theta_2 \frac{CFO_{it}}{TA_{it}} + \theta_3 \frac{CFO_{it+1}}{TA_{it}} + \theta_4 DCFO_{it} + \theta_5 DCFO_{it} \frac{CFO_{it}}{TA_{it}} + \varepsilon_{1it}$$
(1)

where subscripts *i* and *t* denote the firm *i* and year *t*. CAC_{it} represents current accruals and is measured by liabilities and non-cash based asset ($CAC_{it} = \Delta CA_{it} - \Delta CL_{it} - \Delta CASH_{it} + \Delta STDEBT_{it}$, where ΔCA_{it} is the variation of current assets; ΔCL_{it} is the variation of current liabilities; $\Delta CASH_{it}$ is the variation of cash and equivalents; $\Delta STDEBT_{it}$ is the variation of short and current long-term debt); TA_{it} is the average total assets at year *t* and *t*-1 of firm *i*; CFO_{it} denotes the cash flows from operations and is calculated by the difference of net income and current accruals ($CFO_{it} = NI_{it} - CAC_{it}$, where NI_{it} is the net income). CFO_{it-1} is the operation cash flows at year *t*-1 and CFO_{it+1} is the operation cash flows at year *t* + 1; $DCFO_{it}$ is a binary variable which takes the value of one if ($CFO_{it}/TA_{it} - CFO_{it-1}/TA_{it}$) < 0, and zero otherwise.

Dechow and Dichev (2002) argue that the residuals in Equation (1), ε_{1it} , are recognized as the discretionary accruals. The absolute value of residuals ε_{1it} or $(|A_RES_{it}|)$ is acknowledged as the proxy for earnings management. Accordingly, the higher absolute value of the residuals $(|A_RES_{it}|)$ or the greater earnings management corresponds to a lower earnings quality.

In the second step, we use the below regression model to investigate whether foreign shareholders play an active role in firm's earnings management:

$$|A_{\mathcal{R}}ES_{it}| = \delta_0 + \delta_1 FOR_{it} + \delta_2 SIZE_{it} + \delta_3 LEV_{it} + \delta_4 ROA_{it} + \delta_5 AG_{it} + \delta_6 LOSS_{it} + \varepsilon_{2it}$$
(2)

where $|A_RES_{it}|$ is the discretionary accruals, measured by the absolute value of the residuals in Equation (1); FOR_{it} is the percentage of total equity owned by foreigners to total outstanding shares; $SIZE_{it}$ is the natural logarithm of market capitalization; LEV_{it} is total debt to total assets ratio; ROA_{it} is net income to total assets ratio; AG_{it} is the growth of total asset; $LOSS_{it}$ is the binary variable which get value of one if the net income is negative, and zero otherwise.

The coefficient δ_1 captures the statistical significance of the association between international shareholdings and earnings management. The negative (positive) value of coefficient δ_1 suggests that higher ownership of foreigner will lead to the better (lower) quality of earnings.

Furthermore, we employ several firm characteristics as control variables since accruals are likely affected by some factors other than the ownership. Considering to firm size (*SIZE*), Watts and Zimmerman (1986) and Moses (1987) anticipate that large firms are more likely to engage earnings manipulation. However, Ashari, Koh, Tan, and Wong (1994) and Atik (2009) argue that large firms are generally examined by the equity researchers and outside investors, therefore they might have less space to manipulate or manage income. Because the relation between firm size and earnings management is not consistent, we simply expect that there is association between firm size and discretionary accruals.

The leverage degree may have effect on the firm's earnings management activities. DeFond and Jiambalvo (1994) report that high firms are likely to engage more earnings management activities. Nevertheless, high debt companies may be closely overseen by the lenders, hence, they are less likely to adopt aggressive earnings management practices (Chung, Firth, & Kim, 2002; Park & Shin, 2004). We include leverage (*LEV*) to control for the impact of debt effect.

Furthermore, we incorporate return on asset as the proxy of profitability (*ROA*). As argued by Orlitzky, Schmidt, and Rynes (2003) and Chen and Yuan (2004), firms with lower rate of return on asset have lesser earnings quality. Similarly, Ferreira and Matos (2008) document that greater ownership by foreigners is associated with higher operating performance. In light of the above discussion, we expect a positive signal for the association between earnings quality and firm's profitability.

Growth opportunity (AG) is included as the control variable and is measured by the annual total asset growth. Chan, Ferguson, Simunic, and Stokes (2001) and Lui (2003) argue that market participants might provide a higher valuation to firms with good growth opportunities because

these firms tend to have more internal information about the growth prospects. Thus, internal management is likely to reveal positive information through financial reports in which earnings have been manipulated to indicate the firm's profitable prospects (Healy & Palepu, 2003). Consistent with the extant studies, we predict that manager in higher growth firms may engage more discretionary accruals management.

Finally, following Ben-Nasr et al. (2015) and Francis and Wang (2008), we include *LOSS* to control for the economic losses. Previous research addresses that if firms post negative earnings, they have the incentive to manipulate the financial earnings (Callen, Robb, & Segal, 2008; Lang & Lundholm, 1993).

4.2.2. Earnings persistence

Drawing on prior literature, this study adopts earnings persistence as a measurement for quality of reported earnings (Ali, Chen, & Radhakrishnan, 2007; Ben-Nasr et al., 2015; Deng et al., 2017; Francis et al., 2004). Accordingly, we employ earnings persistence as the second proxy for earnings quality to examine the correlation between foreign shareholdings and the quality of reported earnings.

The current study adopts the earnings persistence model from Sloan (1996) to assess the influence of the foreign holdings on earnings persistence:

$$NI_{it} = \beta_0 + \beta_1 NI_{it-1} + \beta_2 NI_{it-1} * FOR_{it} + \beta_3 SIZE_{it} + \varepsilon_{3it}$$
(3)

where NI_{it} is the net earnings at year *t* divided by averaged total assets at year *t*; NI_{it-1} is the net earnings at year *t*-1 divided by averaged total assets at year *t*-1. The other variables are defined earlier.

In Equation (3), the coefficient β_1 measures the earnings persistence. The higher value of β_1 indicates the better quality of earnings. The coefficient β_2 captures the influence of foreign shareholders on the persistence of earnings.

The inclusion of control variables in this model allows us to provide more comprehensive analysis. Lev (1983) reports that firm size is positively relative with earnings persistence. Similarly, Ali et al. (2007) and Ben-Nasr et al. (2015) document that firm size also impact on earnings persistence. However, Li, Abeysekera, and Ma (2014) document that small firms are associated with higher profit persistence. Therefore, we have no directional prediction on the correlation between firm size and earnings persistence.

4.2.3. Earnings informativeness

The market value of listed companies captures the information on reported earnings. Accordingly, firm's net income can be employed to explore the relation between financial reporting quality and equity returns. Particularly, the changes in stock price are implied to be reflected changes in reporting earnings, suggesting that investors embed informativeness of earnings into the stock price. Following the prior studies (Ben-Nasr et al., 2015; Deng et al., 2017; Fan & Wong, 2002; Wang, Wong, & Xia, 2008), we utilize the below regression to understand the impact of foreign ownership on informativeness of earnings:

$$CR_{it} = \alpha_0 + \alpha_1 EPS_{it} + \alpha_2 EPS_{it} * FOR_{it} + \alpha_3 SIZE_{it} + \alpha_4 MTB_{it} + \varepsilon_{4it}$$
(4)

where CR_{it} is measured as the cumulative 12-month stock returns; EPS_{it} is the earnings per share; MTB_{it} denotes the ratio of market value to book value; the other variables are defined earlier.

The coefficient α_1 measures the magnitude of earnings in stock returns. The coefficient α_2 measures differential earnings informativeness in relative to the level of foreign ownership. If the coefficient α_2 is positive, it implies that higher informative of earnings is associated with the larger proportion of foreign stake holdings.

The current study employs firm size (*SIZE*) to control for size effect. As argued by Atiase (1985) and Freeman (1987), larger firms have higher earning-stock return sensitivity than smaller firms. In contrast, Fombrun and Shanley (1990) and Verschoor (2005) document that smaller companies tend to excessively expose their performance to attract the attention of market participants. Therefore, the earnings-equity return relation in small firms might be more sensitive than in large firms.

In addition, we incorporate the market to book ratio to proxy for firm's growth opportunities (*MTB*). Collins and Kothari (1989) document that firm's growth is associated with future earnings. The higher market to book ratio leads to the higher earnings growth, and the stronger earnings–equity returns relation.

4.3. Endogeneity concern

There is a primary concern that ownership variable is recognized as an endogenous variable (Batten & Vo, 2015; Chung & Zhang, 2011; Giannetti & Simonov, 2006; Leuz, Lins, & Warnock, 2010). The endogeneity concern might arise because foreign investors prefer to invest in firms with high earnings quality. Specifically, Leuz et al. (2010) report that international investors will not invest in firms which have potential problems of corporate governance. Likewise, Chung and Zhang (2011) and Giannetti and Simonov (2006) offer evidence that foreign capital is flowed more to well-governed companies. Recently, Batten and Vo (2015) document that foreign investors tend to avoid investing into weak financial management firms where local investors have informational advantages.

Wooldridge (2006) argues that the least squares regression method might be biased and inconsistent estimators if the endogeneity is present in testing model. The author suggests the use of two-stage least squares (2SLS) estimation to overcome this problem. Specifically, in the first stage, the endogenous variable is regressed with the instrumental variables. In the second stage, the main models are estimated for the left-hand variables based on coefficient values of the explanatory variables.

In this paper, we adopt listing period as the instrument variable to enhance the robustness of the analyses. We employ listing period as the instrument variable because in our sample, foreign ownership is associated with listing period whereas three dimensions of earnings quality seem have no correlation with listing period.^{(3),(4)}

5. Analysis of the relation between foreign ownership and earnings quality

In this subsection, we use different estimation models to explore the effect of foreign ownership on the quality of earnings. More specifically, the study analyzes the correlation between foreign shareholdings and earnings management, earnings persistence and earnings informativeness in the context of Vietnam equity market.

We briefly discuss our ordinary least squares (OLS) results and then we present our results from the 2SLS estimation method.

5.1. Descriptive statistics

Table 3 contains summary statistics for the dependent, independent variables and firm-level variables used in the regressions. It appears that, on the average, foreign ownership ratio is 12%. This is comparable to data from Ben-Nasr et al. (2015) who report 12.5% average foreign shareholdings of 45 countries. This statistic is higher than 8.1% of average foreign ownership in China (Firth et al., 2007) and 7.5% in Ghana (Abor & Biekpe, 2007). We denote that the foreign ownership restriction in Vietnam stock market has recently removed since 2015. As the result, the foreign holdings are widely dispersed from the mean value of 5.6% versus the highest owning 49%. Regarding firm characteristics variables, we find that the ratio of return on total assets and leverage ratio has wide dispersion as the average value is around of 8% and 11% while the highest value reaches to more than 80%. We also observe that the average debt rate is at the level of 25%, interpreting that debt does not heavily weigh on Vietnam-listed firms. As shown in Table 3, the average value of market-to-book is 1.27 which is greater than 1, implying the stock prices exceed the book value of listed firms in Vietnam stock market. This statistic also suggests that Vietnamese institutions have strong growth prospect.

Table 3. Sum	mary descriptive	stati stics			
Variables	Mean	Median	Maximum	Minimum	SD
FOR	0.12	0.052	0.49	0.00	0.1435
A_RES	0.06	0.03	1.17	0.00	0.18
NI	0.08	0.06	0.95	-0.59	0.10
CR	0.05	0.00	2.71	-1.23	0.47
LEV	0.25	0.24	0.88	0.00	0.20
LOSS	0.07	0.00	1.00	0.00	0.25
MTB	1.27	0.95	25.45	-14.71	1.36
ROA	0.08	0.06	0.78	-1.58	0.09
SIZE	13.07	13.30	41.72	00.00	2.88
AG	0.22	0.10	8.14	-0.68	0.54
EPS	2,133.88	1,758.83	95,542.00	-21,602.05	3,332.53

Notes: This table reports the summary descriptive statistics of 245 Vietnamese non-financial companies listed on the Ho Chi Minh City Stock Exchange from 2007 to 2015. Source: Bloomberg. *FOR* is the percentage of total equity owned by foreigners to total outstanding shares; *|A_RES|* is the absolute value of discretionary accruals; *NI* is the net earnings; *CR* denotes the stock return and measured as cumulative 12-month stock returns; *LEV* is total debt to total assets ratio; *ROA* is net income to total assets ratio; *AG* is the growth of total asset; *SIZE* is the natural logarithm of market capitalization; *LOSS* is the binary variable which get value of one if the net income is negative, and zero otherwise; *MTB* denotes the ratio of market value to book value; *EPS* is the earnings per share.

High correlation between independent variables might lead to multicollinearity problem. Therefore, we use the Spearman correlation coefficients in Table 4 to identify the multicollinearity problem among the independent variables in this research. As reported in this table, all the correlation coefficients are under 0.8. Following Kennedy (2003), we can conclude that there is no problem of multicollinearity.

5.2. Analysis of the relation between foreign ownership and earnings management

In this section, we investigate the association between the foreign ownership in listed firms and the level of earnings management. Firstly, Table 5 presents the baseline result of the direct relationship between foreign shareholdings and discretionary accruals. The ordinary least square estimation suggests that there is no clear evidence of the association between the two variables in the whole sample, but it shows that higher foreign ownership leads to lower level of earnings management in large firms.

Secondly, Table 6 provides the results from the 2SLS estimation for the analysis of the relation between foreign ownership and discretionary accruals. The negative and significant coefficient on the association between international shareholders and firm's discretionary accruals (-0.1162) indicates that greater foreign shareholdings are aligned with less earnings management in their invested firms. This finding is consistent with previous studies (Ayadi, 2014; Ben-Nasr et al., 2015; Francis & Wang, 2008; Park & Park, 2004; Velury & Jenkins, 2006).

A potential explanation for the negative relationship between the two variables in Vietnam setting is that the greater presence of foreign institutional investors over foreign individual investors. Foreign shareholders with strong knowledge of finance and management might be actively in supervising their portfolio firm's managerial behavior (Almazan, Hartzell, & Starks, 2005). Consequently, higher stock proportion held by foreign shareholders results in an improvement in the earnings quality of Vietnamese listed firms.

Using the mean value of firm size as the benchmark, we partition our sample into two groups, including a group of large companies and a group of small companies. Large companies are the ones which have the value of firm size exceeding the mean value and otherwise. We observe that in the Vietnamese large companies, managers tend to manipulate the earnings with the presence of foreign owners while in the small firms, the earnings management through discretionary accruals is likely to

Table 4. Cor	Table 4. Correlation matrix	i,									
	FOR	A_RES	IN	CR	LEV	ross	MTB	ROA	SIZE	ЪĞ	EPS
FOR	1										
A_RES	0.01	1									
IN	0.25***	0.19***	4								
CR	0.09***	0.09***	0.20***	1							
LEV	-0.16***	0.11***	-0.1***	-0.06**	1						
ross	-0.07***	-0.19**	-0,48***	-0.14*	0.10***	1					
MTB	0.21***	0.10***	0.39***	0.17***	-0.13***	-0.16^{***}	1				
ROA	0.24***	0.19***	1.00***	0.20***	-0.20***	-0.48***	0.39***	1			
SIZE	0.06***	0.06***	0.08***	0.08***	0.14***	-0.03	0.13***	0.08***	1		
AG _{it}	0.00***	0.46***	0.34***	0.07***	0.03***	-0.16^{***}	0.18***	0.34***	0.05***	1	
EPS	0.30***	0.15***	0.29***	0.36***	-0.19***	-0.49***	0.56***	0.80***	0.12***	0.22***	1
Notes: This tab. Bloomberg. FO. averaged total total asset; SIZ value to book v	Notes: This table shows the Spearman correlation matrix of the sample of Vietnamese non-financial companies listed on the Ho Chi Minh City Stock Exchange from 2007 to 2015. Data are collected from Bloomberg. FOR is the percentage of total equity owned by foreigners to total outstanding shares; A.RES/ is the absolute value of discretionary accruals; NI is the net earnings at year t divided by averaged total assets; CR denotes the stock return and measured as cumulative 12-month stock returns; LEV is total debt to total assets ratio; ROA is net income to total assets ratio; Af is the growth of total asset; SIZE is the natural logarithm of market capitalization; LOSS is the binary variable which get value of one if the net income is negative, and zero otherwise; MTB denotes the ratio of market value to book value; EPS is the earnings per share.	irman correlation ge of total equity is the stock returr igarithm of marke iarnings per share	matrix of the san / owned by foreig n and measured a et capitalization; <i>1</i> 3.	nple of Vietname gners to total ou is cumulative 12- LOSS is the binary	se non-financial tstanding shares month stock retu y variable which	companies listed s; A_RES/ is the al urns; LEV is total (get value of one	on the Ho Chi Mi bsolute value of debt to total asse if the net incom	nh City Stock Exc discretionary acc its ratio; ROA is ne s is negative, and	nange from 2007 ruals; NI is the r et income to tota I zero otherwise;	sample of Vietnamese non-financial companies listed on the Ho Chi Minh City Stock Exchange from 2007 to 2015. Data are collected from preigners to total outstanding shares; A.RES/ is the absolute value of discretionary accruals, NI is the net earnings at year t divided by ed as cumulative 12-month stock returns; LEV is total debt to total assets ratio; ROA is net income to total assets ratio; AG is the growth of on; LOSS is the binary variable which get value of one if the net income is negative, and zero otherwise; MTB denotes the ratio of market	e collected from ear t divided by is the growth of ratio of market

	respectively.
	evel,
	1% [
	and 1%
	_
	è 10%, 5%
	e at the
	e at
-	significance
	statistical
	nt the
	represei
	*** re
	and
	*, *, a
	-

discretionary accruals									
Variable	Total sam	ple A_RES		mpanies ZE) A_RES		Small companies (SMALLSIZE) A_RES			
	Coeff.	t-stat.	Coeff.	t-stat.	Coeff.	t-stat.			
FOR	-0.0084	-0.8380	-0.0732***	-2.8265	-0.0138	-0.3428			
SIZE	0.0006	1.0998	-0.0009	-0.3384	0.0012	1.2963			
LEV	-0.0233***	-2.6743	-0.0464	-1.9945	-0.0905***	-3.3672			
ROA	0.1344***	6.2335	0.2816***	6.1143	-0.2399***	-5.9655			
AG	0.0084	1.5299	-0.0051	-0.7350	0.0089	0.8046			
LOSS	0.1032***	15.9465	0.0686***	8.7252	0.0673***	6.4145			
Intercept	0.0366***	5.2496	0.0684	1.9124	0.0701***	5.4705			
R ²	0.1	945	0.5995		0.6571				
No. of Obs	11	19	63	30	48	39			

Notes: This table shows the results of OLS estimator for sample of Vietnamese non-financial companies listed on the Ho Chi Minh City Stock Exchange over the period from 2007 to 2015. *FOR* is the percentage of total equity owned by foreigners to total outstanding shares; $|A_RES|$ is the absolute value of discretionary accruals; *LEV* is total debt to total assets ratio; *ROA* is net income to total assets ratio; *AG* is the growth of total asset; *SIZE* is the natural logarithm of market capitalization; *LOSS* is the binary variable which get value of one if the net income is negative, and zero otherwise. Model (2): $|A_RES_{lt}| = \delta_0 + \delta_1 FOR_{lt} + \delta_2 SIZE_{lt} + \delta_3 LEV_{lt} + \delta_4 ROA_{lt} + \delta_5 AG_{lt} + \delta_6 LOSS_{lt} + \epsilon_{2lt}$.

*, **, and *** represent the statistical significance at the 10%, 5%, and 1% level, respectively.

be unrelated to the presence of international investors. This finding supports the size hypothesis that large firms are liable to manipulate reporting earnings (Moses, 1987; Watts & Zimmerman, 1986).

Regarding the effect of profitability on the relation of foreign shareholdings and earnings management, we find that firm's profitability (*ROA*) is positively associated with discretionary accruals. This result suggests that firms with high profitability are resulted to lower earnings quality. In this sense, the findings pronounce the characteristic of Vietnamese capital market that lower profitable firms are more likely to control the earnings number to attract the foreign investors. This finding is inconsistent with Orlitzky et al. (2003) and Chen and Yuan (2004) who report a positive association between *ROA* and firms' earnings quality.

5.3. Analysis of the relation between foreign ownership and earnings persistence

We firstly present the result of ordinary least squares estimator in Table 7 which shows the association between foreign ownership and the persistence of earnings. We find that there is no clear evidence of the relationship between the level of foreign shareholdings and the degree of earnings persistence.

In the next step, the results from 2SLS estimation for the analysis of the impact of the foreign investors' holdings on earnings persistence are reported in Table 8. The value of the coefficient on the influence of foreign shareholdings and earnings persistence is 0.2896, suggesting that the presence of foreigners significantly and positively influences the corporate earnings persistence. This evidence is in line with the findings from existing studies (Ali et al., 2007; Ben-Nasr et al., 2015; Francis et al., 2004).

In addition, we observe that foreign shareholders play positive role in enhancing the persistence of accounting earnings in large firms group. This result is consistent with finding of prior studies (Ali et al., 2007; Ben-Nasr et al., 2015; Lev, 1983).

5.4. Analysis of the relation between foreign ownership and earnings informativeness

Table 9 provides the ordinary least squares estimates of the association between the level of foreign holdings and the informativeness of earnings. The baseline results reveal that in both large firms and small firms, foreign investors have no clear effect on the earnings informativeness.

LEV

ROA

AG

LOSS

Year

dummv

-/+

-/+

+

+

-0.0265*

0.2396***

0.0013

0.1007***

Yes

-1.9474

8.7889

0.1908

13.8266

Table 6. Two-stage least squares estimation of the relation between foreign ownership and
discretionary accruals

			Par	nel A Fi	rst-Stag	ge Le	ast Squa	′es	
Variable	Total	Total sampleFOR			Large companies (LARGESIZE)FOR				ompanies SIZE)FOR
	Coeff.	t-sta	t.	Co	eff.	t·	-stat.	Coeff.	t-stat.
SIZE	0.0064**	* 2.963	57	0.03	41***	3	.9849	0.0042	1.8181
LEV	-0.0788**	** -3.326	53	-0.07	795**	-2	2.1843	-0.0977***	-2.7545
ROA	0.0249	0.754	9	0.0	475	1	.1212	-0.0696	-1.2465
AG	-0.0080	-1.297	-1.2974		186**	-2	2.2746	-0.0035	-0.2606
LOSS	0.0105	1.051	1.0519		167	1	.1852	-0.0009	-0.0637
LIST_PER	0.0224**	* 6.178	6.1787		57***	4	.7337	0.0095	1.6995
Intercept	0.0416	1.553	1.5535		40***	-2	2.8681	0.0641**	2.3525
R ²		0.8180	8180		0.83	350		0.8281	
No. of Obs		1316	1316			64		5	52
		Panel B S	econo	l-Stage	e Least S	Squa	res (2SLS)	
Variable		Total sam	Total sample A_RES				ompanies ZE) A_RES		companies SIZE) A_RES
	Expected	Coeff.	t-s	stat.	Coef	ff.	t-stat.	Coeff.	t-stat.
FOR	-	-0.1162***	-3.	0021	-0.235	2***	-2.8029	-0.0770	-1.1264
SIZE	-/+	0.0017	1.0	5614	0.014	6**	2.3656	-0.0009	-0.6633

Industry dummy		Yes	Yes	Yes					
R ²		0.1458	0.0969	0.2072					
No. of Obs	o. of Obs 852 464 388								
Minh City Stock outstanding sl income to toto	<pre>K Exchange over hares; A_RES is al assets ratio; Al</pre>	the period from 2007 to 2015. Fo the absolute value of discretion G is the growth of total asset; SI2	le of Vietnamese non-financial (OR is the percentage of total equi hary accruals; LEV is total debt t ZE is the natural logarithm of ma gative, and zero otherwise. Instri	ity owned by foreigners to total o total assets ratio; <i>ROA</i> is net arket capitalization; <i>LOSS</i> is the					

-0.0292

0.4136***

-0.0001

0.0938***

-1.0132

6.8231

-0.0072

5.4357

Yes

-0.0248

0.1202

0.0004

0.1098***

-0.9454

0.9412

0.0232

5.2498

Yes

(LIST_PER). Model (2): $|A_{RES_{it}}| = \delta_0 + \delta_1 FOR_{it} + \delta_2 SIZE_{it} + \delta_3 LEV_{it} + \delta_4 ROA_{it} + \delta_5 AG_{it} + \delta_6 LOSS_{it} + \varepsilon_{2it}$.

*, **, and *** represent the statistical significance at the 10%, 5%, and 1% level, respectively.

To deal with endogeneity problem as mentioned in previous section, we use 2SLS in this analysis. Table 10 summarizes the results from 2SLS estimation for the analysis of the relation between foreign ownership and earnings informativeness on Vietnam stock market. The coefficient on the correlation between foreign shareholders and earnings informativeness is 0.0012 which suggests that foreign holders have a positive impact on earnings informativeness. The presence of foreign shareholders is beneficial for the company on the grounds that their high requirements on accounting practices lead to the improvement of earnings information. The findings suggest that Vietnamese institutions with a higher percentage of foreign holdings have higher informative accounting earnings. The results also support the evidence from prior studies (Ayadi, 2014; Ben-Nasr et al., 2015; Fan & Wong, 2002; Wang, 2006).

Table 7. Ordinary least squares estimation of the relation between foreign ownership and earnings persistence

Variable	Total so	ImpleNI		ompanies SIZE)NI	Small companies (SMALLSIZE)NI			
	Coeff.	t-stat.	Coeff.	t-stat.	Coeff.	t-stat.		
NI(-1)	0.2024***	5.7709	0.05910	1.22018	0.1558***	2.7492		
NI(-1)*FOR	0.0683	0.4177	0.31730	1.56857	-0.1363	-0.4101		
SIZE	0.0005	0.5363	0.00259	0.62442	0.0013	1.0012		
Intercept	0.0497***	4.2361	0.02940	0.50065	0.0422***	2.7419		
R ²	0.6519		0.6916		0.6892			
No. of Obs	15	30	89	98	63	632		

Notes: This table provides the results of OLS estimation for panel sample of Vietnamese non-financial companies listed on the Ho Chi Minh City Stock Exchange over the period from 2007 to 2015. FOR is the percentage of total equity owned by foreigners to total outstanding shares; N_{lt} is the net earnings at year t divided by averaged total assets at year t; N_{lt-1} is the net earnings at year t_1 divided by averaged total assets at year t_1; SIZE is the natural logarithm of market capitalization. Model (3): $N_{lt} = \theta_0 + \theta_1 N_{lt-1} + \theta_2 N_{lt-1} * FOR_{lt} + \theta_3 SIZE_{lt} + \epsilon_{3t}$.

*, **, and *** represent the statistical significance at the 10%, 5%, and 1% level, respectively.

Table 8. Two-stage least squares estimation of the relation between foreign ownership and earnings persistence

	Panel A First-Stage Least Squares								
Variable	Total sa	mpleFOR	-	ompanies SIZE)FOR	Small companies (SMALLSIZE)FOR				
	Coeff.	t-stat.	Coeff.	t-stat.	Coeff.	t-stat.			
NI(-1)	0.1867***	5.9894	0.2119***	4.2750	0.1531***	3.0120			
SIZE	0.0065***	3.4056	0.0467***	5.9899	0.0039*	1.7892			
LIST_PER	0.0584***	7.1111	0.0716***	6.0308	0.0460***	3.5437			
Intercept	-0.0447	-1.6185	-0.6206***	-5.4589	-0.0168	-0.5364			
R ²	0.8726		0.8933		0.8611				
No. of Obs	13	03	751		552				

Panel B Second-Stage Least Squares (2SLS)

Variable		Total sampleNI		Large companies (LARGESIZE)NI		Small companies (SMALLSIZE)NI	
	Expected	Coeff.	t-stat.	Coeff.	t-stat.	Coeff.	t-stat.
NI(-1)	+	0.6546***	19.3340	0.6105***	11.7851	0.7151***	15.4761
NI(-1)*FOR	+	0.2896***	2.6009	0.3064*	1.8939	0.1865	0.9995
SIZE	+/-	0.0004	0.4386	0.0097***	2.8686	0.0003	0.2304
Year dummy		Yes		Yes		Yes	
Industry dummy		Yes		Yes		Yes	
R ²		0.5603		0.5749		0.5604	
No. of Obs		865		470		395	

Notes: This table provides the results of 2SLS estimation for panel sample of Vietnamese non-financial companies listed on the Ho Chi Minh City Stock Exchange over the period from 2007 to 2015. *FOR* is the percentage of total equity owned by foreigners to total outstanding shares; N_{lt} is the net earnings at year t divided by averaged total assets at year t; N_{lt-1} is the net earnings at year t-1 divided by averaged total assets at year t-1; *SIZE* is the natural logarithm of market capitalization. Instrument variable is listing period (LIST_PER). Model (3): $N_{lt} = \theta_0 + \theta_1 N_{lt-1} + \theta_2 N_{lt-1} * FOR_{it} + \theta_3 SIZE_{it} + \epsilon_{3it}$.

*, **, and *** represent the statistical significance at the 10%, 5%, and 1% level, respectively.

earnings informativeness									
Variable	Total sampleCR		-	ompanies SIZE)CR	Small companies (SMALLSIZE)CR				
	Coeff.	t-stat.	Coeff.	t-stat.	Est. Coeff.	t-stat.			
EPS	0.0001 ***	4.4812	0.0000	0.3016	0.0001***	8.9944			
EPS*FOR	0.0000	-0.8233	0.0000	0.6291	-0.0001	-1.1376			
SIZE	-0.0053 -0.5448		-0.0816	-1.5539	-0.0012	-0.1282			
MTB	0.5314***	25.4070	0.6439***	21.7168	0.1447***	4.7156			
Intercept	-0.5802***	-4.4407	0.3520	0.4783	-0.2760***	-2.5389			
R ²	0.4483		0.5420		0.4045				
No. of Obs	16	80	10	03	667				

Table 9. Ordinary least squares estimation of the relation between foreign ownership and earnings informativeness

Notes: This table shows the results using OLS estimation for panel sample of Vietnamese non-financial companies listed on the Ho Chi Minh City Stock Exchange over the period from 2007 to 2015. FOR is the ratio of total equity owned by foreigners to total outstanding shares (in percentage); CR denotes the stock return and measured as cumulative 12-month stock returns; SIZE is the natural logarithm of market capitalization; MTB denotes the ratio of market value to book value; EPS is the earnings per share. Model (4) : $CR_{it} = \alpha_0 + \alpha_1 EPS_{it} + \alpha_2 EPS_{it} * FOR_{it} + \alpha_3 SIZE_{it} + \alpha_4 MTB_{it} + \epsilon_{4it}$.

*, **, and *** represent the statistical significance at the 10%, 5%, and 1% level, respectively.

Furthermore, we find that foreign shareholders positively influence the relationship between earnings and return in large firms and this impact is not clear in small firms. This finding is consistent with Atiase (1985) and Freeman (1987) who report that the earning-return relation is more sensitive in large companies than in small companies.

6. Conclusion

The central objective throughout this paper is to determine the nature role of foreign shareholders in monitoring the discretion in local firm's financial disclosures. To achieve our goal, we analyze the impact of foreign ownership on the different attributes of the earnings quality. Specifically, this research investigates the correlation between foreign ownership and three proxies of earnings quality, namely earnings management, earnings persistence, and earnings informativeness.

The results of multivariate estimations show that foreign shareholders serve as a positive role in enhancing corporate's earnings quality. Particularly, our analysis provides evidence of the negative relationship between foreign ownership and earnings management, suggesting that an increase in foreign holdings percentage results in a decrease in discretionary accruals. Furthermore, our findings show significant evidence of the correlation between foreign owners and earnings persistence. Finally, we find the existence of a significant positive association between the international investors and the earnings informativeness, translating an increase of the percentage of capital held by the foreign investors results in a higher informativeness of earnings. All in all, these findings allow us to report the vital role of foreign shareholders in enhancing the quality of earnings in Vietnamese listed firms.

Our research provides insightful implications for both market participants and policy makers. Initially, the results from the multivariate analysis show that foreign investors positively influence domestic companies' governance and promote the quality of firm's reported earnings. Therefore, equity researchers might consider ownership structure as an important determinant of Vietnamese companies' corporate governance. Furthermore, developing policies to attract foreign capital flow is among priority subset in Vietnam's economic reforms. The findings in this analysis assist policy makers to increase the awareness and comprehensive understanding of the role of foreign ownership in Vietnam capital market.

In this paper, we pronounce the importance of sole foreign ownership role while the other ownership structure types have not been included, for example, state ownership, managerial

Table 10. The analysis of the relation between foreign ownership and earnings informativeness

Panel A First-Stage Least Squares									
Variable	Total sampleFOR		-	mpanies IZE)FOR	Small companies (SMALLSIZE)FOR				
	Coeff.	t-stat.	Coeff.	t-stat.	Est. Coeff.	t-stat.			
EPS	0.0000***	4.3677	0.0000***	4.3446	0.0000***	2.1446			
SIZE	0.0083	1.2838	0.0484***	4.3030	-0.0091	-1.3013			
MTB	0.0202*** 2.2424		0.0111	1.6153	0.0115	1.2105			
LIST_PER	0.0502***	5.7578	0.0472***	4.6540	0.0406***	3.0378			
Intercept	-0.0907	-1.0773	-0.6403***	-4.1998	0.1294	1.4811			
R ²	0.1520		0.2487		0.0799				
No. of Obs	1365		795		570				

Panel B Second-Stage Least Squares (2SLS)

Variable	Expected	Total sampleCR		Large companies (LARGESIZE)CR		Small companies (SMALLSIZE)CR		
		Coeff.	t-stat.	Coeff.	t-stat.	Est. Coeff.	t-stat.	
EPS		-0.0002***	-2.2668	-0.0004***	-2.5017	0.0001*	1.9267	
EPS*FOR	+	0.0012***	2.7465	0.0017***	2.6837	0.0003	0.7665	
SIZE	+/-	-0.0948***	-2.0407	-0.5231***	-2.6642	0.0071	0.6208	
MTB	+	0.6279***	2.6159	0.7673***	3.3265	0.1213	1.2536	
Year dummy		Yes		Yes		Yes		
Industry dummy		Ye	es	Yes		Yes		
R ²		0.2	0.2040		0.2257		0.4095	
No. of Obs		925		513		412		

Notes: This table shows the results using 2SLS estimation for panel sample of Vietnamese non-financial companies listed on the Ho Chi Minh City Stock Exchange over the period from 2007 to 2015. *FOR* is the ratio of total equity owned by foreigners to total outstanding shares (in percentage); *CR* denotes the stock return and measured as cumulative 12-month stock returns; *SIZE* is the natural logarithm of market capitalization; *MTB* denotes the ratio of market value to book value; *EPS* is the earnings per share. Instrument variable is listing period. Model $(4) : CR_{it} = \alpha_0 + \alpha_1 EPS_{it} + \alpha_2 EPS_{it} + \alpha_3 SIZE_{it} + \alpha_4 MTB_{it} + \varepsilon_{4it}$.

*, **, and *** represent the statistical significance at the 10%, 5%, and 1% level, respectively.

ownership which may have a possible impact on the earnings quality of Vietnamese listed companies. In terms of directions for future research, we suggest a further study could assess the effects of these ownership types on earnings quality.

Acknowledgements

This paper was presented at the INFINITI Conference on International Finance Asia-Pacific, at the University of Economics Ho Chi Minh City, Vietnam on 7–8 December 2016. We are especially grateful for many insightful comments and helpful suggestions on earlier version of the present paper from Hong Bo, SOAS University of London, UK. The helpful comments of other conference participants are also gratefully acknowledged.

(1) Retrieved 11 January 2017 from http://data.world bank.org/indicator/BN.KLT.PTXL.CD

(2)Retrieved 11 January 2017 from http://doimoi doanhnghiep.chinhphu.vn/so-lieu/

(3) In the untabulated results, foreign ownership is highly correlated to listing period whereas the three

attributes of earnings quality (earnings management, earnings persistence, earnings informativeness) are not associated with listing period.

(4) Validity of the instrument variable: *F*-statistic values in the first stage of earnings management, earnings persistence, and earnings informativeness models are 33.15, 38.17, and 50.56, respectively. These *F*-statistic values are well above ten, that says my instrument pass the rule of thumb (Stock, Wright, & Yogo, 2002).

Funding

The authors received no direct funding for this research.

Author details Xuan Vinh Vo¹ E-mail: vxvinh@yahoo.com E-mail: vxvinh@yahoo.com ORCID ID: http://orcid.org/0000-0002-4725-6914 Thi Kim Huong Chu²

- E-mail: huong.chu@maybank-kimeng.com.vn ¹ Institute of Business Research and CFVG Ho Chi Minh City, University of Economics Ho Chi Minh City, 59C Nguyen Dinh Chieu Street, District 3, Ho Chi Minh City, Vietnam.
- ² Institute of Business Research, University of Economics Ho Chi Minh City, 59C Nguyen Dinh Chieu Street, District 3, Ho Chi Minh City, Vietnam.

Citation information

Cite this article as: Do foreign shareholders improve corporate earnings quality in emerging markets? Evidence from Vietnam, Xuan Vinh Vo & Thi Kim Huong Chu, *Cogent Economics & Finance* (2019), 7: 1698940.

References

- Abor, J., & Biekpe, N. (2007). Corporate governance, ownership structure and performance of SMEs in Ghana: Implications for financing opportunities. Corporate Governance: the International Journal of Business in Society, 7(3), 288–300. doi:10.1108/ 14720700710756562
- Ali, A., Chen, T.-Y., & Radhakrishnan, S. (2007). Corporate disclosures by family firms. *Journal of Accounting and Economics*, 44(1–2), 238–286. doi:10.1016/j. iacceco.2007.01.006
- Almazan, A., Hartzell, J. C., & Starks, L. T. (2005). Active institutional shareholders and costs of monitoring: Evidence from executive compensation. *Financial Management*, 34(4), 5–34. doi:10.1111/fima.2005.34.issue-4
- Ashari, N., Koh, H. C., Tan, S. L., & Wong, W. H. (1994). Factors affecting income smoothing among listed companies in Singapore. Accounting and Business Research, 24(96), 291–301. doi:10.1080/ 00014788.1994.9729488
- Atiase, R. K. (1985). Predisclosure information, firm capitalization, and security price behavior around earnings announcements. *Journal of Accounting Research*, 23(1), 21–36. doi:10.2307/2490905
- Atik, A. (2009). Detecting income-smoothing behaviors of Turkish listed companies through empirical tests using discretionary accounting changes. *Critical Perspectives on Accounting*, 20(5), 591–613. doi:10.1016/j.cpa.2008.01.003
- Ayadi, W. M. (2014). The relationship between ownership structure and earnings quality in the French context. International Journal of Accounting and Economics Studies, 2(2), 80–87.
- Aydin, N., Sayim, M., & Yalama, A. (2007). Foreign ownership and firm performance: Evidence from Turkey. International Research Journal of Finance and Economics, 11(2007), 103–111.
- Baba, N. (2009). Increased presence of foreign investors and dividend policy of Japanese firms. *Pacific-Basin Finance Journal*, 17(2), 163–174. doi:10.1016/j. pacfin.2008.04.001
- Batten, J. A., & Vo, X. V. (2015). Foreign ownership in emerging stock markets. Journal of Multinational Financial Management, 32–33, 15–24. doi:10.1016/j. mulfin.2015.05.001
- Ben-Nasr, H., Boubakri, N., & Cosset, J.-C. (2009). Ownership Structure and Earnings Quality: Evidence from Newly Privatized Firms, Working Paper-HEC. Canada, pp. 1–53.
- Ben-Nasr, H., Boubakri, N., & Cosset, J.-C. (2015). Earnings quality in privatized firms: The role of state and foreign

owners. Journal of Accounting and Public Policy, 34(4), 392–416. doi:10.1016/j.jaccpubpol.2014.12.003

- Callen, J. L., Robb, S. W., & Segal, D. (2008). Revenue manipulation and restatements by loss firms. Auditing: A Journal of Practice & Theory, 27(2), 1–29. doi:10.2308/aud.2008.27.2.1
- Cao, L., Du, Y., & Hansen, J. Ø. (2017). Foreign institutional investors and dividend policy: Evidence from China. *International Business Review*, 26, 816–827. Available online 13 February 2017. doi: 10.1016/j. ibusrev.2017.02.001.
- Chan, D., Ferguson, A., Simunic, D., & Stokes, D. (2001). A spatial analysis and test of oligopolistic competition in the market for audit services, Working paper. Hong Kong University of Science and Technology, University of Technology, Sydney, and University of British Columbia.
- Chen, G., Firth, M., Gao, D. N., & Rui, O. M. (2006). Ownership structure, corporate governance, and fraud: Evidence from China. Journal of Corporate Finance, 12(3), 424–448. doi:10.1016/j. jcorpfin.2005.09.002
- Chen, K. C., & Yuan, H. (2004). Earnings management and capital resource allocation: Evidence from China's accounting-based regulation of rights issues. *The Accounting Review*, 79(3), 645–665. doi:10.2308/ accr.2004.79.3.645
- Choi, J. J., Lam, K. C., Sami, H., & Zhou, H. (2013). Foreign ownership and information asymmetry. Asia-Pacific Journal of Financial Studies, 42(2), 141–166. doi:10.1111/ajfs.12010
- Chung, K. H., & Zhang, H. (2011). Corporate governance and institutional ownership. *Journal of Financial and Quantitative Analysis*, 46(1), 247–273. doi:10.1017/ S0022109010000682
- Chung, R., Firth, M., & Kim, J.-B. (2002). Institutional monitoring and opportunistic earnings management. *Journal of Corporate Finance*, 8(1), 29–48. doi:10.1016/S0929-1199(01)00039-6
- Collins, D. W., & Kothari, S. (1989). An analysis of intertemporal and cross-sectional determinants of earnings response coefficients. *Journal of Accounting and Economics*, 11(2–3), 143–181. doi:10.1016/0165-4101(89)90004-9
- Cuomo, F., Mallin, C., & Zattoni, A. (2016). Corporate governance codes: A review and research agenda. Corporate Governance: an International Review, 24(3), 222–241. doi:10.1111/corg.12148
- de Sousa, E. F., & Galdi, F. C. (2016). The relationship between equity ownership concentration and earnings quality: Evidence from Brazil. *Revista De Administração*, 51(4), 331–343. doi:10.1016/j. rausp.2016.07.006
- Dechow, P., Ge, W., & Schrand, C. (2010). Understanding earnings quality: A review of the proxies, their determinants and their consequences. *Journal of Accounting and Economics*, 50(2–3), 344–401. doi:10.1016/j.jacceco.2010.09.001
- Dechow, P. M., & Dichev, I. D. (2002). The quality of accruals and earnings: The role of accrual estimation errors. The Accounting Review, 77(s-1), 35–59. doi:10.2308/accr.2002.77.s-1.35
- DeFond, M. L., & Jiambalvo, J. (1994). Debt covenant violation and manipulation of accruals. Journal of Accounting and Economics, 17(1–2), 145–176. doi:10.1016/0165-4101(94)90008-6
- Degeorge, F., Ding, Y., Jeanjean, T., & Stolowy, H. (2013). Analyst coverage, earnings management and financial development: An international study. *Journal of Accounting and Public Policy*, 32(1), 1–25. doi:10.1016/j.jaccpubpol.2012.10.003

- Deng, L., Li, S., & Liao, M. (2017). Dividends and earnings quality: Evidence from China. International Review of Economics & Finance, 48, 255–268. doi:10.1016/j. iref.2016.12.011
- Desender, K. A., Aguilera, R. V., Lópezpuertas-Lamy, M., & Crespi, R. (2014). A clash of governance logics: Foreign ownership and board monitoring. *Strategic Management Journal*, 37(2), 349–369. doi:10.1002/ smj.2344
- Dvořák, T. (2005). Do domestic investors have an information advantage? Evidence from Indonesia. The Journal of Finance, 60(2), 817–839. doi:10.1111/ j.1540-6261.2005.00747.x
- Fan, J. P., & Wong, T. J. (2002). Corporate ownership structure and the informativeness of accounting earnings in East Asia. *Journal of Accounting and Economics*, 33(3), 401–425. doi:10.1016/S0165-4101 (02)00047-2
- Ferreira, M. A., & Matos, P. (2008). The colors of investors' money: The role of institutional investors around the world. Journal of Financial Economics, 88(3), 499– 533. doi:10.1016/j.jfineco.2007.07.003
- Firth, M., Fung, P. M., & Rui, O. M. (2007). Ownership, twotier board structure, and the informativeness of earnings–Evidence from China. *Journal of Accounting* and Public Policy, 26(4), 463–496. doi:10.1016/j. jaccpubpol.2007.05.004
- Fombrun, C., & Shanley, M. (1990). What's in a name? Reputation building and corporate strategy. Academy of Management Journal, 33(2), 233–258.
- Francis, J., LaFond, R., Olsson, P. M., & Schipper, K. (2004). Costs of equity and earnings attributes. *The Accounting Review*, *79*(4), 967–1010. doi:10.2308/ accr.2004.79.4.967
- Francis, J. R., & Wang, D. (2008). The joint effect of investor protection and Big 4 audits on earnings quality around the world. *Contemporary Accounting Research*, 25(1), 157–191. doi:10.1506/car.25.1.6
- Freeman, R. N. (1987). The association between accounting earnings and security returns for large and small firms. *Journal of Accounting and Economics*, 9(2), 195–228. doi:10.1016/0165-4101(87)90005-X
- Giannetti, M., & Simonov, A. (2006). Which investors fear expropriation? Evidence from investors' portfolio choices. The Journal of Finance, 61(3), 1507–1547. doi:10.1111/j.1540-6261.2006.00879.x
- Guo, J., Huang, P., Zhang, Y., & Zhou, N. (2015). Foreign ownership and real earnings management: Evidence from Japan. Journal of International Accounting Research, 14(2), 185–213. doi:10.2308/jiar-51274
- Haxhi, I., & Aguilera, R. V. (2017). An institutional configurational approach to cross-national diversity in corporate governance. *Journal of Management Studies*, 54(3), 261–303. doi:10.1111/joms.2017.54. issue-3
- He, W., Ng, L., Zaiats, N., & Zhang, B. (2017). Dividend policy and earnings management across countries. *Journal of Corporate Finance*, 42, 267–286. doi:10.1016/j.jcorpfin.2016.11.014
- Healy, P., & Palepu, K. G. (2003). How the quest for efficiency corroded the market. *Harvard Business Review*, 76–85.
- Hribar, P., & Craig Nichols, D. (2007). The use of unsigned earnings quality measures in tests of earnings management. *Journal of Accounting Research*, 45(5), 1017–1053. doi:10.1111/joar.2007.45.issue-5
- Huang, R. D., & Shiu, C. Y. (2009). Local effects of foreign ownership in an emerging financial market: Evidence from qualified foreign institutional investors in Taiwan. *Financial Management*, 38(3), 567–602. doi:10.1111/j.1755-053X.2009.01048.x

- Jung, K., & Kwon, S. Y. (2002). Ownership structure and earnings informativeness: Evidence from Korea. The International Journal of Accounting, 37(3), 301–325. doi:10.1016/S0020-7063(02)00173-5
- Kang, S., Sul, W., & Kim, S. (2010). Impact of foreign institutional investors on dividend policy in korea: A stock market perspective. *Journal of Financial Management and Analysis*, 23(1), 10.
- Karolyi, G. A., & Stulz, R. M. (2003). Are financial assets priced locally or globally? *Handbook of the Economics* of Finance, 1, 975–1020.
- Katz, S. P. (2009). Earnings quality and ownership structure: The role of private equity sponsors. The Accounting Review, 84(3), 623–658. doi:10.2308/ accr.2009.84.3.623
- Kennedy, P. (2003). A Guide to Econometrics (Vol. 1).
- Khalil, M., & Ozkan, A. (2016). Board independence, audit quality and earnings management: Evidence from Egypt. Journal of Emerging Market Finance, 15(1), 84– 118. doi:10.1177/0972652715623701
- Klai, N., & Omri, A. (2011). Corporate governance and financial reporting quality: The case of Tunisian firms. *International Business Research*, 4(1), 158.
- La Porta, R., & Lopez-de-Silanes, F. (1998). Capital markets and legal institutions. In *Beyond the Washington* consensus: Institutions matter (pp. 73–92).
- La Porta, R., Lopez-de-Silanes, F., Shleifer, A., & Vishny, R. (2000). Investor protection and corporate governance. Journal of Financial Economics, 58(1–2), 3–27. doi:10.1016/S0304-405X(00)00065-9
- Lang, M., & Lundholm, R. (1993). Cross-sectional determinants of analyst ratings of corporate disclosures. *Journal of Accounting Research*, 31(2), 246–271. doi:10.2307/2491273
- Leuz, C., Lins, K. V., & Warnock, F. E. (2010). Do foreigners invest less in poorly governed firms? *Review of Financial Studies*, 23(3), 3245–3285. doi:10.1093/rfs/hhn089.ra
- Lev, B. (1983). Some economic determinants of timeseries properties of earnings. *Journal of Accounting and Economics*, 5(1), 31–48. doi:10.1016/0165-4101 (83)90004-6
- Li, F., Abeysekera, I., & Ma, S. (2014). The effect of financial status on earnings quality of Chinese-listed firms. *Journal of Asia-Pacific Business*, 15(1), 4–26. doi:10.1080/10599231.2014.872963
- Li, Y., & Giles, D. E. (2015). Modelling volatility spillover effects between developed stock markets and asian emerging stock markets. *International Journal of Finance & Economics*, 20(2), 155–177. doi:10.1002/ijfe.v20.2
- Lo, K. (2008). Earnings management and earnings quality. Journal of Accounting and Economics, 45(2–3), 350– 357. doi:10.1016/j.jacceco.2007.08.002
- Lui, D. W. (2003). When Accruals Meet Growth: Do Analysts' Forecasts Fully Reflect the Future earnings Implications of Accruals and Growth? Working paper, London Business School.
- McKnight, P. J., & Weir, C. (2009). Agency costs, corporate governance mechanisms and ownership structure in large UK publicly quoted companies: A panel data analysis. The Quarterly Review of Economics and Finance, 49(2), 139–158. doi:10.1016/j. aref.2007.09.008
- Mehrani, S., Moradi, M., & Eskandar, H. (2017). Institutional ownership type and earnings quality: Evidence from Iran. *Emerging Markets Finance and Trade*, 53(1), 54–73. doi:10.1080/1540496X.2016.1145114
- Moses, O. D. (1987). Income smoothing and incentives: Empirical tests using accounting changes. *The Accounting Review*, 62(2), 358–377.
- Myers, J. N., Myers, L. A., & Omer, T. C. (2003). Exploring the term of the auditor-client relationship and the

quality of earnings: A case for mandatory auditor rotation? *The Accounting Review*, 78(3), 779–799. doi:10.2308/accr.2003.78.3.779

- Orlitzky, M., Schmidt, F. L., & Rynes, S. L. (2003). Corporate social and financial performance: A meta-analysis. Organization Studies, 24(3), 403–441. doi:10.1177/ 0170840603024003910
- Park, M. S., & Park, T. (2004). Insider sales and earnings management. Journal of Accounting and Public Policy, 23(5), 381–411. doi:10.1016/j. jaccpubpol.2004.07.003
- Park, Y. W., & Shin, -H.-H. (2004). Board composition and earnings management in Canada. Journal of Corporate Finance, 10(3), 431–457. doi:10.1016/ S0929-1199(03)00025-7
- Richardson, S. A., Sloan, R. G., Soliman, M. T., & Tuna, A. (2001). Information in accruals about the quality of earnings (July 2001). Retrieved from SSRN: https:// ssrn.com/abstract=278308
- Sánchez-Ballesta, J. P., & García-Meca, E. (2007). Ownership structure, discretionary accruals and the informativeness of earnings. Corporate Governance: An International Review, 15(4), 677–691. doi:10.1111/ corg.2007.15.issue-4
- Schipper, K., & Vincent, L. (2003). Earnings quality. Accounting Horizons, 17, 97–110. doi:10.2308/ acch.2003.17.s-1.97
- Singh, M., & Davidson III, W. N. (2003). Agency costs, ownership structure and corporate governance mechanisms. Journal of Banking & Finance, 27(5), 793–816. doi:10.1016/S0378-4266(01)00260-6
- Sloan, R. G. (1996). Do stock prices fully reflect information in accruals and cash flows about future earnings? The Acccounting Review, 71(3), 289–315.
- Stock, J. H., Wright, J. H., & Yogo, M. (2002). A survey of weak instruments and weak identification in generalized method of moments. *Journal of Business & Economic Statistics*, 20(4), 518–529. doi:10.1198/ 073500102288618658
- Velury, U., & Jenkins, D. S. (2006). Institutional ownership and the quality of earnings. *Journal of Business*

Research, 59(9), 1043–1051. doi:10.1016/j. jbusres.2006.05.001

- Verschoor, C. C. (2005). Is there financial value in corporate values? Strategic Finance, 87(1), 17.
- Vo, X. V. (2015). Foreign ownership and stock return volatility–Evidence from Vietnam. Journal of Multinational Financial Management, 30, 101–109. doi:10.1016/j.mulfin.2015.03.004
- Vo, X. V. (2016). Foreign investors and corporate risk taking behavior in an emerging market. *Finance Research Letters*, 18, 273–277. doi:10.1016/j.frl.2016.04.027
- Wang, D. (2006). Founding family ownership and earnings quality. Journal of Accounting Research, 44(3), 619– 656. doi:10.1111/j.1475-679X.2006.00213.x
- Wang, Q., Wong, T.-J., & Xia, L. (2008). State ownership, the institutional environment, and auditor choice: Evidence from China. *Journal of Accounting and Economics*, 46(1), 112–134. doi:10.1016/j. jacceco.2008.04.001
- Watts, R. L., & Zimmerman, J. L. (1986). Positive accounting theory. Prentice-Hall Inc., 1986. Retrieved from SSRN: https://ssrn.com/abstract=928677
- Wild, J. J. (1996). The audit committee and earnings quality. Journal of Accounting, Auditing & Finance, 11(2), 247–276. doi:10.1177/ 0148558X9601100206
- World Bank. 2013. Vietnam Report on the Observance of Standards and Codes (ROSC): Corporate governance country assessment. Retrieved from http://siteresources.worldbank.org/ FINANCIALSECTOR/Resources/WBG-742_CG_ROSC_ Vietnam FINAL_web.pdf
- Xiao, J. Z., Yang, H., & Chow, C. W. (2004). The determinants and characteristics of voluntary Internetbased disclosures by listed Chinese companies. *Journal of Accounting and Public Policy*, 23(3), 191–225. doi:10.1016/j.jaccpubpol.2004.04.002
- Xu, X., & Wang, Y. (1999). Ownership structure and corporate governance in Chinese stock companies. *China Economic Review*, 10(1), 75–98. doi:10.1016/ S1043-951X(99)00006-1



© 2020 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license.

You are free to:

Share — copy and redistribute the material in any medium or format. Adapt — remix, transform, and build upon the material for any purpose, even commercially. The licensor cannot revoke these freedoms as long as you follow the license terms. Under the following terms: Attribution — You must give appropriate credit, provide a link to the license, and indicate if changes were made. You may do so in any reasonable manner, but not in any way that suggests the licensor endorses you or your use. No additional restrictions You may not apply legal terms or technological measures that legally restrict others from doing anything the license permits.

Cogent Economics & Finance (ISSN: 2332-2039) is published by Cogent OA, part of Taylor & Francis Group. Publishing with Cogent OA ensures:

- Immediate, universal access to your article on publication
- High visibility and discoverability via the Cogent OA website as well as Taylor & Francis Online
- Download and citation statistics for your article
- Rapid online publication
- Input from, and dialog with, expert editors and editorial boards
- Retention of full copyright of your article
- Guaranteed legacy preservation of your article
- Discounts and waivers for authors in developing regions

Submit your manuscript to a Cogent OA journal at www.CogentOA.com